Author’s response to reviews

Title: Feeding styles of caregivers of children 6-23 months in Derache special district, Southern Ethiopia

Authors:

Tseganeh Amsalu (tseganeh2009@yahoo.com)
Mekitie Wondafrash (mwondafrash@yahoo.com)
Mirkuzie Woldie (mirkuzie@yahoo.com)

Version: 2 Date: 27 July 2011

Author’s response to reviews: see over
Re: Submission of a revised version of the manuscript

Dear Sir/Madam,

On the manuscript entitled “Feeding styles of caregivers of children 6-23 months in Derache special district, Southern Ethiopia”, we have made a change of the software used for analysis. SPSS was used initially but later on we used Stata version 11 for re-analysis of the whole data. Hence, under the subtitle of “Data analysis” the software used has been changed from SPSS version 16 (SPSS Inc, Chicago IL USA) to Stata version 11 (StataCorp, College Station, TX, USA).

The title “DISCUSSION” in upper case was changed to lower case “Discussion” Moreover, some formatting has been made.

Best regards,

Mekitie Wondafrash (MD, DFSN)

Corresponding author